Author’s response to reviews

Title: Acceptability of screening for pregnancy intention in general practice: a population survey of people of reproductive age

Authors:

Karin Hammarberg (karin.hammarberg@monash.edu)
Julie Hassard (jhassard@varta.org.au)
Renee de Silva (RdeSilva@varta.org.au)
Louise Johnson (ljohnson@varta.org.au)

Version: 2 Date: 23 Jan 2020

Author’s response to reviews:

Dear Dr Gaines,

Thank you for your email regarding our manuscript. In response to your requests we have:

1) Added that consent was obtained in writing online.
2) Added a list of abbreviations after the conclusion.
3) Uploaded a ‘clean’ copy of the manuscript.

We hope our paper is now suitable for publication and again thank the reviewers for their constructive comments.

Yours sincerely,

Karin Hammarberg (on behalf of the authors)